Author's response to reviews

Title: Epidemiology of leisure-time physical activity in socio-demographic, lifestyle and psychological characteristics of men and women in Greece: the ATTICA Study

Authors:

Demosthenes Panagiotakos (d.b.panagiotakos@usa.net)
Christos Pitsavos (d.b.panagiotakos@usa.net)
Yannis Lentzas (d.b.panagiotakos@usa.net)
Christodoulos Stefanadis (d.b.panagiotakos@usa.net)

Version: 3 Date: 12 April 2005

Author's response to reviews:

Dear Editor,

Thank you so much for your interest in our work.

We have made all the editorial changes.

D B Panagiotakos